Author's response to reviews

Title: Assessment of fall-related self-efficacy and activity avoidance in people with Parkinson's disease

Authors:

Maria H Nilsson (Maria_H.Nilsson@med.lu.se)
Anna-Maria Drake (Anna_Maria.Drake@med.lu.se)
Peter Hagell (Peter.Hagell@med.lu.se)

Version: 4 Date: 15 October 2010

Author's response to reviews: see over
Dear Editor,
We have now revised our manuscript (Title: Assessment of fall-related self-efficacy and activity avoidance in people with Parkinson's disease) in accordance with your requests (see below). We hope that these revisions are sufficient for a publication in BMC Geriatrics. Please contact the corresponding author otherwise.

Kind regards,
Maria H Nilsson

Revisions

Major revisions
- The list of authors is now written in accordance with the submission system.
- Conclusions: We have changed “Discussion” to “Discussion and Conclusions” as suggested.

Minor revision
- All highlighting is removed.
- “Participants and methods” is renamed “Methods”
- We have also checked our manuscript for any typographical errors.